Author's response to reviews

Title: Comparison of semi-quantitative and quantitative culture techniques for the diagnosis of catheter-related infections in newborns and molecular typing of isolated microorganisms

Authors:

Danilo FM Riboli (danilo.riboli@ibb.unesp.br)
João C Lyra (jc.lyra@uol.com.br)
Eliane P Silva (eliane_bio10@yahoo.com.br)
Luisa L Valadão (luisaleit@gmail.com)
Maria R Bentlin (mbentlin@fmb.unesp.br)
José E Corrente (jecorren@ibb.unesp.br)
Lígia MSS Rugolo (ligiasr@fmb.unesp.br)
Maria LRS Cunha (cunhamlr@ibb.unesp.br)

Version: 2 Date: 5 August 2013

Author's response to reviews:

Protocol number of institutional Ethics Committee aprovation was added on materials and methods. We wold like to transfer the submitted manuscript to BMC Infectious Diseases as suggested.